Author's response to reviews

Title: Clinical Evaluation Study of the German Network of Disorders of Sex Development (DSD)/Intersexuality: study design, description of the study population, and data quality

Authors:

Anke Lux (Anke.Lux@med.ovgu.de)
Siegfried Kropf (Siegfried.Kropf@med.ovgu.de)
Eva Kleinemeier (kleinemeier@paedia.ukl.mu-luebeck.de)
Martina Juergensen (juergensen@paedia.ukl.mu-luebeck.de)
Ute Thyen (thyen@paedia.ukl.mu-luebeck.de)

Version: 2 Date: 24 March 2009

Author's response to reviews:

Melissa Norton, MD
Editor-in-chief
BMC Public Health

Submission “Clinical Evaluation Study of the German Network of Disorders of Sex Development (DSD)/Intersexuality: study design, description of the study population, and data quality“

Luebeck, 24/03/2009

Dear Melissa Norton,

We would like to submit the revised version of the study protocol paper: Clinical Evaluation Study of the German Network of Disorders of Sex Development (DSD)/Intersexuality: study design, description of the study population, and data quality.

1. We structured the abstract according your guidelines.

2. We include a 'Competing interests' section between the Conclusions and Authors' contributions.

3. Within the description of the study design (before the description of the instruments) we included a power calculation to determine the sample size of your study.

We hope that the revisions are sufficient. If there are further questions do not hesitate to contact us.

Best regards
Ute Thyen, MD